Author’s response to reviews

Title: Syphilitic hepatitis: A case report and review of the literature

Authors:

Jiaofeng Huang (huangjiaofeng@fjmu.edu.cn)
Su Lin (sumer5129@fjmu.edu.cn)
Mingfang Wang (wangmingfang@fjmu.edu.cn)
Bo Wan (wblu@163.com)
Yueyong Zhu (zhuyueyong@fjmu.edu.cn)

Version: 3  Date: 02 Nov 2019

Author’s response to reviews:

Dear Abraham,

We would like to thank you for thoroughly reviewing our manuscript entitled ‘Syphilitic hepatitis: case report and review of the literature’ (BMGE-D-19-00220R1). We acknowledge the reviewers’ comments and constructive suggestions very much.

Sincerely,

Yueyong Zhu, M.D.
Associated Professor of Medicine
Liver Disease Research Center
The First Affiliated Hospital of Fujian Medical University
Email: zhuyueyong@fjmu.edu.cn

Here are our point-by-point responses:

Editor Comments:

1. Role of Funders
Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

Thank you for your suggestion. We have changed the funding part to “This work was supported by the Medical Project of Fujian Province (2016-CX-33), Fujian Province Health Youth Research Project
(2019-1-37) and Fujian Medical University Sailing Fund Project (2018QH1047). The funders had no role in study design, conduct and outcome of the manuscript. No additional external funding was received for this study.”

2. Authors’ contributions
Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript in its current state.
Thank you for your suggestion. The Author contributions was “HJF did the data collection and wrote the primary draft. LS did the study design. WMF and WB were involved in manuscript preparation. LS and ZYY contributed to reporting edits and the idea for the Clinical Picture. All authors have read and approved the manuscript in its current state.”

3. Figure 1
We note that a figure (Figure 1) is cited in the body of text and in the Figure legends, however no figure or image was supplied in the text nor as a supplementary file. Please ensure all figures that should be included are sufficiently provided in or with a revised version of the manuscript.
Sorry for not uploading Figure 1 in the prior revision. The picture has been re-uploaded in this revision, please check.

4. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Thank you! We have modified the article and uploaded the final version as required.